Author's response to reviews

Title: The utility of exome sequencing for genetic diagnosis in a familial microcephaly epilepsy syndrome

Authors:

Laura M. McDonell (mcdonell.laura@gmail.com)
Jodi Warman Chardon (jwarman@cheo.on.ca)
Jeremy Schwartzentruber (jeremy.schwartzentruber@mail.mcgill.ca)
Denise Foster (dfoster@algomapublichealth.com)
Chandree L Beaulieu (cbeaulieu@cheo.on.ca)
FORGE Canada Consortium FORGE Canada Consortium (care4rare@cheo.on.ca)
Janet Marcadier (jmarcadier@cheo.on.ca)
Dennis E Bulman (dbulma@cheo.on.ca)
Kym M Boycott (kboycott@cheo.on.ca)

Version: 3 Date: 17 January 2014

Author's response to reviews:

We are happy to hear about the recommended acceptance. To clarify, Janet Marcadier is not to be included as an author as she did not participate in experiments, manuscript preparation or revision. We apologize for this oversight, we had copied and pasted a list of email addresses and hers was inadvertently included.